Author's response to reviews

Title: Modeling the variations in pediatric respiratory syncytial virus seasonal epidemics

Authors:

Molly Leecaster (molly.leecaster@hsc.utah.edu)
Per Gesteland (per.gesteland@hsc.utah.edu)
Tom Greene (tom.greene@hsc.utah.edu)
Nephi Walton (nephi.walton@utah.edu)
Adi Gundlapalli (adi.gundlapalli@hsc.utah.edu)
Robert Rolfs (rrolfs@utah.gov)
Carrie Byington (carrie.byington@hsc.utah.edu)
Matthew Samore (matthew.samore@hsc.utah.edu)

Version: 4 Date: 18 April 2011

Author's response to reviews:

Dear Editors,

Other than minor grammatical corrections, we have made the following revisions, as per your request. We 1) included a Discussion section, 2) provided functioning links to websites in reference numbers 13 and 14, 3) cited Figure 4 in the text, 5) removed all highlighting and tracks, 6) changed the 'et al.' to author names in references, and 7) removed visible vertical lines from the tables.

We appreciate your time and efforts.

Molly Leecaster